Author’s response to reviews

Title: Measurement Error and Reliability of three available 3D Superimposition Methods in Growing Patients

Authors:
Cecilia Ponce-Garcia (poncegar@ualberta.ca)
Antonio Carlos de Oliveira Ruellas (aruellas@umich.edu)
Lucia Helena Soares Cevidanes (luciaclev@umich.edu)
Carlos Flores-Mir (cf1@ualberta.ca)
Jason Carey (jason.carey@ualberta.ca)
Manuel Lagravere-Vich (manuel@ualberta.ca)

Version: 2 Date: 28 Nov 2019

Author’s response to reviews:

Reviewer #1: The comments originally made have NOT been addressed. The two 'sentences' are:

1. Page 17 Line 4-11 (quite clear)
   
   Known as problematic areas due to growth that occurs during childhood and adolescence at the sphenoo-occipital synchondrosis and the remodeling that occurs at the surfaces of the occipital and posterior sphenoid bones including the posterior wall of sella, which goes under resorption till late teens. Author’s response: Thanks for your detailed comment - the sentence has been fully edited and is ready for your review.

2. My second comment relates to what now appears on Page 18 Lines 25- 29
   
   The possible effect of the segmentation process and the different software programs used for the full superimposition as well as the landmark identification as a source of measurement error in 3D radiographic imaging. Author’s response: Thanks for your detailed comment - the sentence has been fully edited and is ready for your review.